

GOPEN ACCESS

Citation: Bell TM, Facemire P, Bearss JJ, Raymond JL, Chapman J, Zeng X, et al. (2024) Smallpox lesion characterization in placebo-treated and tecovirimat-treated macaques using traditional and novel methods. PLoS Pathog 20(2): e1012007. https://doi.org/10.1371/journal.ppat.1012007

Editor: Antonio Alcamí, Centro de Biologia Molecular Severo Ochoa, SPAIN

Received: June 14, 2023

Accepted: January 26, 2024

Published: February 22, 2024

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: The work described herein was supported by the Defense Threat Reduction Agency (DTRA). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Opinions, interpretations, conclusions, and recommendations are those of the authors and are RESEARCH ARTICLE

Smallpox lesion characterization in placebotreated and tecovirimat-treated macaques using traditional and novel methods

Todd M. Bell¹[•], Paul Facemire¹[•], Jeremy J. Bearss¹, Jo Lynne Raymond¹, Jennifer Chapman¹ⁿ, Xiankun Zeng¹, Joshua D. Shamblin¹, Janice A. Williams¹, Douglas W. Grosenbach², Dennis E. Hruby², Inger K. Damon³, Arthur J. Goff¹, Eric M. Mucker¹

1 U.S. Army Medical Research Institute of Infectious Diseases (USAMRIID), Frederick, Maryland, United States of America, 2 SIGA Technologies, Inc., Corvallis, Oregon, United States of America, 3 Poxvirus and Rabies Branch, Centers for Disease Control and Prevention (CDC) Atlanta, Georgia, United States of America

• These authors contributed equally to this work.

¤ Current address: Labcorp Early Development Laboratories Inc., Chantilly, Virginia, United States of America

* toddbell10@hotmail.com

Abstract

Smallpox was the most rampant infectious disease killer of the 20th century, yet much remains unknown about the pathogenesis of the variola virus. Using archived tissue from a study conducted at the Centers for Disease Control and Prevention we characterized pathology in 18 cynomolgus macagues intravenously infected with the Harper strain of variola virus. Six macaques were placebo-treated controls, six were tecovirimat-treated beginning at 2 days post-infection, and six were tecovirimat-treated beginning at 4 days postinfection. All macaques were treated daily until day 17. Archived tissues were interrogated using immunohistochemistry, in situ hybridization, immunofluorescence, and electron microscopy. Gross lesions in three placebo-treated animals that succumbed to infection primarily consisted of cutaneous vesicles, pustules, or crusts with lymphadenopathy. The only gross lesions noted at the conclusion of the study in the three surviving placebo-treated and the Day 4 treated animals consisted of resolving cutaneous pox lesions. No gross lesions attributable to poxviral infection were present in the Day 2 treated macagues. Histologic lesions in three placebo-treated macaques that succumbed to infection consisted of proliferative and necrotizing dermatitis with intracytoplasmic inclusion bodies and lymphoid depletion. The only notable histologic lesion in the Day 4 treated macaques was resolving dermatitis; no notable lesions were seen in the Day 2 treated macagues. Variola virus was detected in all three placebo-treated animals that succumbed to infection prior to the study's conclusion by all utilized methods (IHC, ISH, IFA, EM). None of the three placebo-treated animals that survived to the end of the study nor the animals in the two tecovirimat treatment groups showed evidence of variola virus by these methods. Our findings further characterize variola lesions in the macague model and describe new molecular methods for variola detection.

not necessarily endorsed by the U.S. Army or the Department of Defense.

Competing interests: DWG and DEH are employed by SIGA Technologies. SIGA Technologies holds commercial interest in tecovirimat and provided tecovirimat for the study referenced. Other authors have no conflict of interest.

Author summary

The worldwide 2022 mpox outbreak renewed research interest in defining and countering threats posed by orthopoxviruses, including smallpox. In the absence of optimized prevention and treatment strategies, natural re-emergence of smallpox or its use as a bioweapon released into unvaccinated populations could result in avoidable illnesses and deaths. Here we applied recently developed techniques to study the relationship between smallpox lesion development and the fate of virus in three groups of animals exposed to smallpox virus: those treated with the FDA-approved antiviral tecovirimat 2 or 4 days after infection and a placebo-treated group. We were able to characterize the development of lesions at gross and microscopic levels and describe new molecular methods for study-ing the distribution and persistence of variola virus (VARV) in tissue samples that were collected from the three treatment groups. These findings can be used to fine-tune the timing of administering tecovirimat, other antivirals, and vaccines, as well as help characterize the performance of in-development orthopoxvirus countermeasures.

Introduction

Smallpox was successfully eradicated worldwide by 1980 through an intensive surveillance, vaccination, and containment program led by the World Health Organization [1–4]. Because of the large-scale success of this unprecedented eradication effort, ongoing vaccination campaigns were discontinued for the general public creating a vacuum of protection against smallpox and other zoonotic poxviral infections.

In the weeks immediately following the 9/11 terrorist attacks, the anthrax letter attacks focused U.S. government attention on national security and existing biodefense gaps, including the gap in poxvirus immunity [5–8]. To address this gap, the U.S. government funded vaccine and antiviral development for biodefense purposes [9–11]. Animal models of smallpox disease as well as next generation vaccines and antivirals were the result [12–17], including tecovirimat (formerly ST-246), an antiviral that is specific for orthopoxviruses. The mechanism of action of this antiviral is inhibition of the VP37 protein preventing proper formation of enveloped virions (EV)[18,19]. It was FDA approved through use of the Animal Rule and was a beneficiary of post-9/11 drug development funding [13,14,20–26].

Using archived tissues from a study conducted at the Centers for Disease Control and Prevention, we endeavored to describe lesions in placebo-treated macaques to further characterize the model and describe lesion mitigation in tecovirimat-treated macaques. We also wanted to determine if viral antigen was detectable 28 days postinfection in either placebo-treated or tecovirimat-treated macaques, indicating viral persistence. Viral persistence amongst other high consequence pathogens has been an active area of research at the U.S. Army Medical Research Institute of Infectious Diseases (USAMRIID) over the past decade [27–32], and with our rich archive of formalin-fixed, paraffin-embedded (FFPE) tissues infected with high consequence pathogens, our goal was to add to the growing database of knowledge regarding viral persistence, particularly in immune privileged sites in this smallpox model.

Much of what we know about smallpox is based on data gathered prior to the eradication of smallpox and the last reported natural human case in 1977. This was prior to the bulk of the biotech revolution, thus limiting our knowledge of the molecular basis of smallpox. Over the past ten years, new and improved orthopoxvirus-specific molecular and immunodiagnostic tools have been developed at USAMRIID to better understand viral pathogenesis, allowing more sensitive detection of viral antigen and viral nucleic acid in tissues.

This study applies some of these new methods to archived FFPE tissues, collected as part of a previous study and stored at USAMRIID, to describe the pathology of placebo-treated versus tecovirimat-treated animals [23]. RNA *in situ* hybridization, immunofluorescence assays, and electron microscopic examination from select target tissues in animals from each treatment group were performed. Results are presented along with observations made during gross necropsies performed on all animals.

Due to the paucity of published pathology data on NHP models of smallpox disease, we endeavored to better characterize the gross lesions and histopathologic findings for this disease model to include a characterization of lesions between placebo-treated and tecovirimat-treated groups. Additionally, we wanted to determine the utility of newly developed molecular, immunodiagnostic, and ultrastructural methods in this characterization and to determine if virus remained in any tissues at the end of the study. We were able to use this arsenal of techniques to further characterize the model and show a lack of virus across all treatment groups and tissues examined.

Results

Gross pathology

Three of six placebo-treated animals survived to the end of the study (28–30 days post-exposure) and three met clinical endpoint criteria between days 13 and 14 and were humanely euthanized (See survival curve from Mucker et al[23]. All 12 tecovirimat-treated NHPs survived exposure to the end of the study. (See <u>Table 1</u>). Skin lesions observed, maximum viremia,

| Table 1. Skin lesions observe | d, maximum viremia | , time to maximum viren | nia, and survival b | y tecovirimat treatment da | ıy. |
|-------------------------------|--------------------|-------------------------|---------------------|----------------------------|-----|
|-------------------------------|--------------------|-------------------------|---------------------|----------------------------|-----|

| Nonhuman Primate Number | Treatment | Skin Le | ions Viremia (gen/mL) | | | Time To Maxir (Day | Outcome* (Day Euthanized) | |
|----------------------------|-----------|---------------------|-----------------------|---------------------|------------------|-----------------------|------------------------------|--------|
| | Day | Maximum Observed | Group Average | Maximum Observed | Group Average | Maximum Viremia | Group Average | |
| 3 | 2 | 655 | 269 | 2.83E+06 | 1.03E+06 | 4 | 3.2 | S(29) |
| 6 | 2 | 46 | | 3.73E+05 | | 4 | | S(28) |
| 8 | 2 | 77 | 1 | 3.11E+04 | | 2 | | S(29) |
| 11 | 2 | 393 | | 1.25E+06 | | 2 | | S (30) |
| 12 | 2 | 131 | | 1.31E+06 | | 4 | | S (28) |
| 18 | 2 | 310 | 1 | 4.07E+05 | | 4 | | S (30) |
| 1 | 4 | 419 | 482 | 1.58E+05 | 4.32E+06 | 2 | 4.7 | S(30) |
| 5 | 4 | 63 | | 2.44E+06 | | 4 | | S(28) |
| 10 | 4 | 335 | | 1.73E+05 | | 4 | | S(28) |
| 14 | 4 | 870 | | 2.12E+07 | | 7 | | S(28) |
| 16 | 4 | 917 | | 7.63E+05 | | 7 | | S(29) |
| 17 | 4 | 286 | 1 | 1.16E+06 | | 4 | | S(28) |
| 2 | Placebo | 1329 | 1479 | 6.47E+06 | 7.68E+06 | 7 | 7 | S(29) |
| 7 | Placebo | 1682 | 1 | 6.48E+06 | | 7 | | S(29) |
| 15 | Placebo | 514 | | 4.21E+05 | 7 | 7 |] | S(29) |
| 4 | Placebo | 1873 | | 6.44E+06 | | 7 |] [| E(14) |
| 9 | Placebo | 1575 | | 9.67E+06 | | 7 |] [| E(14) |
| 13 | Placebo | 1900 |] | 1.66E+07 | 7 | 7 | | E(13) |

* S, Survived until end of study; E, met clinical criteria for euthanasia previous to end of study

Note: End of study necropsies were dispersed over 3 days due to the extreme amount of time and effort it took to do the NHP necropsies and sample collections in BSL-4. Data for the number of skin lesions observed, maximum viremia, time to maximum viremia, and survival is from previously published data (ref [23]).

https://doi.org/10.1371/journal.ppat.1012007.t001

Table 2. Gross findings by tecovirimat treatment group.

| | Nonhuman Primate Number | | | | | | | | | | | | | | | | | |
|---|-------------------------|---------|----|---|---|----|----|-----------------|---|---|---|----|-----------------|----|----|----|---|----|
| | | Placebo | | | | | | Day 2 Treatment | | | | | Day 4 Treatment | | | | | |
| Observed Lesions | 9* | 13* | 4* | 7 | 2 | 15 | 11 | 18 | 3 | 8 | 6 | 12 | 1 | 16 | 14 | 17 | 5 | 10 |
| Cutaneous vesicles, pustules, crusts &/or scars | X | X | X | — | — | _ | — | _ | _ | _ | _ | _ | _ | _ | _ | — | _ | _ |
| Enlarged lymph nodes | | X | X | — | X | — | — | — | _ | Х | _ | — | _ | — | Х | X | _ | _ |
| Pale, raised pleural lesion(s) or pleural adhesions | | X | - | - | - | - | - | _ | Х | Х | _ | _ | _ | - | Х | - | _ | - |
| Mottled red lung lobes | — | _ | X | — | — | — | — | _ | _ | _ | _ | _ | _ | _ | _ | — | _ | _ |
| Oral cavity ulcers | X | X | X | — | — | — | — | _ | _ | _ | _ | _ | _ | _ | _ | — | _ | _ |
| Enlarged spleen | X | X | X | - | X | X | X | _ | Х | Х | _ | _ | Х | - | _ | X | _ | - |
| Esophageal hemorrhagic foci/ulcers | X | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | |
| Gastrointestinal mucosal petechial hemorrhage or erosions | X | X | X | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | |
| Resolving Pox lesions on feet or hands | _ | _ | _ | X | X | X | _ | _ | _ | _ | _ | _ | _ | _ | Х | X | _ | _ |

X = finding noted;— = no finding

*Euazed prior to end of study because study endpoint met; all other animals survived to end of study

https://doi.org/10.1371/journal.ppat.1012007.t002

time to maximum viremia, and survival by tecovirimat treatment day are listed in Table 1 and are from previously published data (ref [23]).

The three NHPs that did not survive to the end of the study had gross lesions consistent with those previously described in the ordinary smallpox model of infection (33). The following gross findings were noted in these three animals (See <u>Table 2</u>): cutaneous vesicles, pustules, crusts and/or scars (Fig 1, includes images of those that met criteria and those that survived



Fig 1. Variola-infected, placebo treated NHPs. Multifocal proliferative and necrotizing dermatitis (pox lesions). A) Facial skin (13 days post infection); B) Abdominal skin, left inner thigh, and scrotum (7 days post infection); C) Glabrous skin, left palm: multifocal proliferative dermatitis (7 days post infection); D) Glabrous skin, left palm: multifocal healed lesions with depigmentation (29 days post infection).

https://doi.org/10.1371/journal.ppat.1012007.g001

| Location | Non | Nonhuman Primate Number | | | | | | | |
|---|-----|-------------------------|---|--|--|--|--|--|--|
| Histologic Findings | 9 | 13 | 4 | | | | | | |
| Haired skin Dermatitis, proliferative and/or necrotizing | 2 | 4 | 2 | | | | | | |
| Tongue Glossitis, proliferative and/or necrotizing | 2 | - | _ | | | | | | |
| Tonsil Tonsillitis +/- Lymphoid changes | - | 2 | 2 | | | | | | |
| Nares Dermatitis, proliferative and/or necrotizing | 3 | - | 3 | | | | | | |
| Lip Dermatitis, proliferative and/or necrotizing | 2 | 2 | 3 | | | | | | |
| Lymph node(s) Lymphadenitis | 4 | 2 | 3 | | | | | | |
| Spleen Lymphoid depletion | 3 | 2 | _ | | | | | | |
| Stomach Gastritis, necrotizing | - | 3 | _ | | | | | | |
| Intestine Enteritis, necrotizing | - | 3 | 4 | | | | | | |

Table 3. Summary of histologic findings for placebo treated NHPs that met study endpoint before end of study.

Severity scores: 1 = Minimal, 2 = Mild, 3 = Moderate, 4 = Severe;— = no finding

NOTE: No significant lesions were seen in the following organs: submandibular salivary gland, larynx, esophagus, trachea, thyroid gland, parathyroid gland, testis, liver, spleen, adrenal glands, pituitary gland, kidneys, urinary bladder, prostate gland, heart, lung, skeletal muscle sciatic nerve, brachial plexus, brain, eyes, and femoral bone marrow.

https://doi.org/10.1371/journal.ppat.1012007.t003

until the end of the study) (3/3); enlarged lymph nodes (3/3); pale, raised pleural lesion(s) or pleural adhesions (2/3); mottled red lung lobes (1/3); ulceration within the oral cavity (3/3); dark red, enlarged spleen (3/3); esophageal hemorrhagic foci/ulcers (1/3); gastrointestinal mucosal petechial hemorrhage or erosions (3/3).

The remaining three placebo-treated animals had gross lesions limited to resolving (healing) poxviral lesions on the feet and/or hands, with 2 also having grossly enlarged spleens and one having enlarged lymph nodes at the study's conclusion. These healing skin lesions were pale, circular, poorly pigmented areas on the palms and footpads. (See <u>Table 2</u> and <u>Fig 1</u>).

Among the 12 tecovirimat-treated NHPs, only 2 treated animals, those from the Day 4 tecovirimat group [14,17] had notable lesions consisting of resolving poxviral lesions of the skin; other findings in these animals were related to an expected immunologic response to infection with enlargement of the lymph nodes and spleen, and non-specific inflammation with respect to the pleural adhesions. The healing skin lesions in these 2 NHPs were consistent with resolving poxviral infection as described above and noted in Table 2.

Histology, immunohistochemistry, In situ hybridization, immunofluorescence, and electron microscopy

In the placebo-treated group, the 3 macaques that succumbed prior to the end of the study had histologic lesions and positive immunohistochemical findings consistent with VARV infection. The histopathology included proliferative and/or necrotizing dermatitis (3/3); proliferative and/ or necrotizing glossitis (1/3); tonsillitis (2/3); proliferative and/or necrotizing dermatitis within the nares (2/3) and lip (3/3); lymphadenitis (3/3); lymphoid depletion within the spleen (2/3); necrotizing gastritis (1/3); and necrotizing enterties (2/3). (See Table 3 and Fig 2).



Fig 2. Variola-infected, placebo treated NHP x days post infection. Haired skin, pox lesions. A: (H&E, 4x) Proliferative and necrotizing dermatitis. B: (IHC, 4x) Viral antigen is present within epithelial cells at the edge of the pox lesion. C: (H&E 20x) Higher magnification of A, Necrotic debris and superficial serocellular crust. D: (IHC, 40x) Viral antigen in epithelial cells. E: (40x/100x) Higher magnification of A, Pyknotic and karyorrhectic (necrotic) debris with intracytoplasmic inclusions (arrowhead; inset, higher magnification of black box) and ballooning degeneration (arrow) as well as surface colonies of bacteria. F: (IHC, 60x) Viral antigen in epithelial cells.

https://doi.org/10.1371/journal.ppat.1012007.g002

All three of these animals had multiple epithelial surfaces that were multifocally IHC positive to include haired skin (3/3), tongue (1/3), lip (2/3), nares (1/3), tonsil (2/3), jejunum (1/3) and ileocecal junction (1/3). See Table 5 and Fig 2 for detailed IHC results.

To further validate the VARV infection, we developed an RNA *in situ* hybridization (ISH) assay to detect VARV-specific transcript B19R, which is deleted in mpox virus (MPXV). This assay specifically detects VARV RNA but not MPXV RNA in respectively infected tissues (Fig 3). Epithelial tissues from the three placebo-treated macaques that did not survive to the end of the study and were positive by IHC were also found to be positive by our newly developed *in situ* hybridization technique designed specifically for variola virus. (Fig 4A and 4B). Additionally, immunofluorescence (IFA) was performed on haired skin of NHP #13 and #4, placebo-treated NHPs that met euthanasia endpoint criteria prior to the end of the study. IFA demonstrates VARV infection primarily infects epithelial cells, but not macrophages (Fig 4C and 4D).

All three surviving placebo-treated animals had histologic lesions within the haired skin consistent with resolving (healing) pox lesions (lymphohistiocytic dermatitis with epidermal



Fig 3. The specificity of variola virus-specific RNA in situ hybridization (ISH). A) variola virus (VARV) RNA was detected in the tongue tissue of a cynomolgus macaque, (placebo-treated #9) that succumbed to intravenous exposure of VARV using RNA ISH. B-C.) VARV RNA was undetectable in the lung tissue of a cynomolgus macaque that succumbed to aerosol exposure MPX virus using VARV-specific ISH (B), in which MPXV RNA was readily detected by MPXV-specific ISH (C).

https://doi.org/10.1371/journal.ppat.1012007.g003



Fig 4. Detection of variola virus infection using molecular techniques. A-B) Variola virus (VARV) antigen (brown in A) and RNA (red in B) were detected in the haired skin of NHP #13, a placebo-treated cynomolgus macaque that succumbed to VARV exposure; immunohistochemistry (IHC) and RNA in situ hybridization (ISH), respectively. C-D) Immunofluorescence staining demonstrates VARV viral antigen (green in C and D) was primarily detected in epithelial cells (red, C) of NHP #4, a placebo-treated cynomolgus macaque, stained by anti-pan-cytokeratin (PanCK) antibody, but not in macrophages stained by anti-C68 antibody (red, D). Nuclei were counterstained blue with 4',6-diamidino-2-phenylindole (DAPI).

https://doi.org/10.1371/journal.ppat.1012007.g004

| Location | Nonhuman Primate Number | | | | | | | | | | | | | | |
|---|-------------------------|---|----|-----------------|----|---|---|---|----|---|-----------------|----|----|----|----|
| Histologic Findings | Placebo | | | Day 2 Treatment | | | | | | | Day 4 Treatment | | | | |
| | 7 | 2 | 15 | 11 | 18 | 3 | 8 | 6 | 12 | 1 | 16 | 14 | 17 | 5 | 10 |
| Haired skin Dermatitis, lymphohistiocytic, with epidermal hyperplasia resolving pox lesion | 2 | 2 | 2 | - | — | _ | — | - | _ | _ | _ | _ | 2 | - | - |
| Nares Dermatitis, lymphohistiocytic, with epidermal hyperplasia resolving pox lesion | — | _ | _ | 2* | — | _ | — | - | _ | _ | _ | _ | — | - | - |
| Lip Dermatitis, lymphohistiocytic, with epidermal hyperplasia resolving pox lesion | — | - | - | _ | — | - | — | - | _ | _ | _ | _ | _ | 1* | - |
| Tonsil Follicular hyperplasia | | 1 | 1 | 1 | — | 1 | 1 | 1 | 1 | _ | _ | 1 | 1 | 1 | 1 |
| Lymph node Follicular hyperplasia +/- sinus histiocytosis | 1 | 1 | 1 | 2 | _ | 1 | 1 | 1 | 1 | 1 | 2 | 1 | 1 | 1 | 1 |
| Spleen White pulp hyperplasia | 1 | 2 | 2 | 2 | 1 | 2 | 2 | 2 | 2 | 2 | 2 | 2 | 2 | 1 | 2 |

Table 4. Summary of histologic findings for placebo and tecovirimat treated NHPs that survived to end of study by tecovirimat treatment group.

*Unclear if this is related to previous poxviral infection; it may be a nonspecific finding.

Severity scores: 1 = Minimal, 2 = Mild, 3 = Moderate, 4 = Severe;— = no finding

NOTE: No significant lesions were seen in the following organs: tongue; submandibular salivary gland, larynx, esophagus, trachea, thyroid gland, parathyroid gland, testis^{**}, liver, adrenal glands, pituitary gland, kidneys, urinary bladder, prostate gland, heart, lung, skeletal muscle sciatic nerve, brachial plexus, stomach, pylorus, duodenum, jejunum, ileum, cecum, proximal and distal colon, brain, eyes, and femoral bone marrow.

** NHP 14: Testis—inflammation, lymphohistiocytic and eosinophilic, multifocal, mild with fibrosis and spermatogenic arrest

https://doi.org/10.1371/journal.ppat.1012007.t004

hyperplasia [Table 4]), in addition to lymphoid hyperplasia. We were unable to detect poxviral antigen using immunohistochemistry and/or *in situ* hybridization in any of the three placebotreated survivors at the end of the study (Table 5).

For animals that were treated with tecovirimat beginning on Day 4, two animals, #14 and #17, had mild inflammation within the haired skin resembling previously described resolving pox lesions, (lymphohistiocytic dermatitis with epidermal hyperplasia) (Table 4). This correlated with the pale, circular areas of skin noted on gross examination. A non-specific finding of unknown clinical significance in a Day 4 treated animal included one primate with mild chronic inflammation in the nares (#5). Similar to the Day 2 treated animals (described in the next paragraph), the Day 4 treated animals exhibited follicular hyperplasia in the tonsil (4/6) and lymph node (6/6), and white pulp hyperplasia in the spleen (6/6), but these animals lacked evidence of active infection with poxvirus using immunohistochemistry, immunofluorescence, *in situ* hybridization, or electron microscopy. In NHP 14 within the testis, there was active production and division of germ cells along the basement membrane of the seminiferous tubules, but the lumina of most seminiferous tubules had few spermatozoa. This indicates an arrest in production of sperm, with a subsequent re-initiation of production. The exact cause is not histologically evident, but systemic infection with VARVand a resultant febrile response followed by convalescence is the most likely explanation for this finding.

For animals that were treated with tecovirimat beginning on Day 2, histologic findings were sparse and most findings were indicative of an immunological response to disease. Histologic findings included follicular hyperplasia in the tonsil (5/6) and lymph node (5/6), and white pulp hyperplasia in the spleen (6/6), findings which are consistent with an appropriate immune response to an infectious agent. Furthermore, these animals lacked evidence of active infection with poxvirus using immunohistochemistry, immunofluorescence, *in situ* hybridization, or electron microscopy. In all groups, pale, raised pleural lesion(s) or pleural adhesions were seen during gross necropsies, but no pathology or associated IHC positivity were seen in

| Location | | Nonhuman Primate Number | | | | | | | | | | | | |
|--------------------|-----|-------------------------|------------------|---|-----|-----|-----|-----|-----|--|--|--|--|--|
| | | Met clinical o | criteria and eut | Survived and euthanized at end of study | | | | | | | | | | |
| | | 9 | 1 | .3 | | 4 | 7 | 2 | 15 | | | | | |
| | IHC | ISH | IHC | ISH | IHC | ISH | IHC | IHC | IHC | | | | | |
| Haired skin | 4+ | / | 4+ | 4+ | 4+ | 4+ | - | - | - | | | | | |
| Tongue | 4+ | 4+ | - | - | - | - | - | - | - | | | | | |
| Tonsil | - | - | 4+ | 4+ | 4+ | 4+ | - | - | - | | | | | |
| Nares | - | - | - | - | 1 | 1 | - | - | - | | | | | |
| Lip | 2+ | 2+ | - | - | 4+ | 4+ | - | - | - | | | | | |
| Lymph Node(s) | - | - | - | - | - | - | - | - | - | | | | | |
| Larynx | 2 | / | - | / | - | / | - | - | - | | | | | |
| Jejunum | - | - | - | - | 1 | - | - | - | - | | | | | |
| Ileocecal Junction | - | - | - | - | 1 | - | - | - | - | | | | | |
| Lung | - | - | - | - | - | - | - | - | - | | | | | |
| Spleen | - | - | - | - | - | - | - | - | - | | | | | |

Table 5. Immunohistochemistry and in situ hybridization findings: Placebo-treated NHPs.

All tissues were stained for IHC; if tissues are not listed, they were negative

/ = ISH data was unavailable

Intensity of staining is graded on the following scale

- = No significant staining

1.....1-10 cells/high power field (hpf)

2....11-20 cells/hpf

3..... 21-40 cells/hpf

4.....>40 cells/hpf

+ = very strong, positive staining

https://doi.org/10.1371/journal.ppat.1012007.t005

any lung tissue of any animals. This points towards a non-specific inflammatory response within all groups.

Electron microscopy results

Haired skin samples from four placebo-treated animals and two non-human primates (NHPs) treated 4 days post-infection with tecovirimat were examined. Abundant VARV are present in the haired skin of three of the four placebo-treated controls euthanized for meeting endpoint criteria (Fig 5). All three control samples of haired skin also exhibit superficial bacteria. The virions and bacteria are juxtaposed on the cell surface and within the epithelial skin layers (Fig 5). All three samples display areas of disrupted cell layers with an invasion of immune cells, to include macrophages, and red blood cells. Some regions show an increase in fibrous material surrounding clusters of viral particles. Three phenotypes, (1) circular immature (Fig 5C', white asterisk), (2) mature virions with bar-bell shaped cores (Fig 5A', 5B', 5C', arrowhead) and (3) bricked shaped envelope mature virions (Fig 5B', black asterisk), are prominent in the viral clusters (Fig 5). The virions measure, on average, 262nm in length and 142nm in width (See Fig 5A').

The placebo-treated control that survived to the study's conclusion, as well as the two animals that were treated 4 days post-exposure, display normal cellular architecture of the haired skin with clearly distinct cell layers. Superficial bacteria are also seen in the surface layers of the haired skin in the infected control animal and one treated survivor. No VARV are seen in any of these samples. Furthermore, an increasing number of stratum corneum layers is seen in all three animals, but no viral particles are present in this expanded layer (Fig 6).



Fig 5. Transmission electron micrographs of VARV infected, placebo-treated NHP haired skin. Panels A, B, C all reveal flaking epidermis and disorganized dermal skin layers in three euthanized variola infected, placebo-treated non-human primates. Panels A', B', and C' are higher magnification images of the boxed regions in panels A, B, and C, respectively, revealing the presence of VARV. Immature (5C', white asterisk) and mature virions are evident. Brick-shaped, enveloped mature (black asterisk) and mature virions with bar-bell shaped cores (arrowheads) are readily visible. Inset in A' shows the dimensions (~ 139nm X 246nm) of one mature virion. Superficial bacteria are also evident (arrows) in panels A and C' (arrows). A', B', C' scalebar = 500nm.

https://doi.org/10.1371/journal.ppat.1012007.g005

Discussion

With the 2022 global emergence of MPXV [34–37], there has been renewed interest in research to better understand orthopoxviruses. Recent studies at USAMRIID and the CDC have shown the efficacy of tecovirimat against MPXV and VARV [13,23,38]. Because of the paucity of data regarding smallpox lesions in NHPs, this retrospective pathologic analysis was done to describe the gross, histologic, and ultrastructural smallpox lesions in both placebo-treated and tecovirimat-treated NHPs. A secondary goal of this analysis was to interrogate archived tissues to determine if virus remained in any tissues in placebo-treated or tecovirimat-treated macaques. Newly developed molecular techniques to detect VARV aided in this determination. Recent data has shown viral persistence in immune privileged sites (gonad) in NHPs infected with MPXV [29], and this discovery warranted review of these tissues in smallpox infected NHPs.

In this study, all NHPs were treated daily until day 17 with either the placebo or tecovirimat, and our goal was to describe the differences in the lesions and viral detection in those lesions between groups. In the placebo treatment group, three animals succumbed to disease prior to the end of the study, and all three had gross, histologic, and immunohistochemical findings consistent with an ongoing VARV infection. We were also able to use newly developed ISH probes to detect VARV, and these specific probes have added utility because they can delineate VARV from MPXV in tissues.

The remaining three placebo-treated animals all had gross and histologic lesions consistent with a resolving poxviral infection. There was no detectable VARV signature via IHC, ISH, IFA, or EM in these 3 placebo-treated end-of-study survivors. This suggests viral clearance is complete by day 28 in the target tissues examined. Unlike other high-consequence viruses that have shown evidence of viral persistence following recovery, VARV did not appear in



Fig 6. Transmission electron micrographs of VARV infected, tecovirimat-treated NHP haired skin. Panels A and B from surviving NHPs reveal more organized epidermis and dermis skin layers in two end-of-study VARV infected, tecovirimat-treated non-human primates. Panels A' and B', are slightly higher magnifications of the epidermis and dermis layers, respectively. No variola virus is evident.

https://doi.org/10.1371/journal.ppat.1012007.g006

resolving lesions through interrogation with our standard and newly developed assays. The overall number of NHPs in this study was very small, and previous studies done in development of this animal model showed a robust amount of variola antigen within the testes during studies that ended on day 11. [33]. Given the relatively small sample number and how rare this finding could be, it remains to be seen if viral antigen would be seen in the testes and surrounding tissues if the overall number of NHPs was increased with additional studies.

For animals treated with tecovirimat beginning at 4 days post-infection, 2/6 animals had enlarged lymph nodes, 2/6 had a dark, enlarged spleen, and 2/6 had lesions consistent with resolving cutaneous pox lesions similar to placebo-treated survivors. Histologically, 6/6 animals had hyperplasia within the lymph nodes and 6/6 had white pulp hyperplasia within the spleen. Follicular hyperplasia was also present within the tonsils of 4/6 animals. These histologic findings are indicative of antigenic stimulation and are expected in animals mounting an appropriate immune response to an infectious agent. No VARV signature was detected in any animal in this group through IHC, ISH, IFA, or EM. Even when tecovirimat was not administered until after the onset of lesions (day 4 post-infection), the NHPs were able to completely clear all remnants of virus. This was also confirmed via electron microscopy.

In NHPs treated with tecovirimat beginning at 2 days post-infection, or prior to lesion onset, there were no specific gross, histologic, or immunohistochemical findings indicating poxviral infection. Grossly, the enlarged lymph nodes (1/6) and dark, enlarged spleen (3/6) correlated histologically to lymphoid hyperplasia in both of these organs, which is a non-

specific response to stimulation from a wide variety of antigens. No variola antigen was detected in the day 2 tecovirimat treatment group via any of our assays in any animal in this group. This indicates treatment with tecovirimat as late as 2 days postinfection results in minimal to no gross or histological lesions, to include no detection of viral protein or viral nucleic acid with the methods described above. The grossly evident pleural adhesions seen across all groups was attributed to non-specific inflammation based on a lack of corresponding histologic findings and a complete absence of IHC positivity in the pleural lining of all examined NHPs.

Conclusion

Overall, tecovirimat ameliorated the typical gross and histologic lesions associated with VARV induced disease in macaques when animals were treated as late as day 4 post-infection. Placebotreated animals experienced a 50% mortality rate, with lesions and viral identification as described in previous studies, demonstrating consistency of this animal model. Additional methods (ISH, IFA) that were optimized for smallpox infection assisted in confirmation of viral presence in the affected tissues. These assays help identify viral protein, viral nucleic acid, and cell types infected. These assays also suggested lack of viral persistence in examined tissues.

Initiation of treatment as late as 4 days post infection, representing the post-lesional model, resulted in 100% survival. In the post-lesion treatment model, animals did have some residual pathologic changes similar to the placebo-treated survivors. This indicates that, while the treatment was successful in eliminating mortality in this group and reducing morbidity, the delayed treatment course did result in development of some minor residual lesions. Treatment administered 2 days postinfection not only resulted in 100% survival but had the additional benefit of preventing any significant pathologic response over the duration of the study. The ability to treat as early as possible, in this case prior to lesion development, can have an impact on disease course and residual pathology.

Some of the limitations of this study include a limited number of animals [18], as well as a single-sex cohort (only males were utilized in these studies). While we did not see any viral persistence due to the limited number of animals in this study, further persistence research should be explored. Viral antigen was found in testes in the Wahl-Jensen study [33], and a larger number of subjects may have resulted in finding virus in the testes of some NHPs in this study.

Mpox, an orthopoxvirus closely related to VARV, has shed a spotlight on the gap in our collective immunity to orthopoxvirues. Understanding the molecular mechanisms of disease for these viruses is of paramount importance. Given the highly transmissible nature of these viruses, any re-emergence of smallpox in now unvaccinated populations is likely to lead to widespread infection before the diagnosis of the index case. Therefore, a better understanding of the pathogenesis of this disease with development of a robust intervention strategy, including the optimization of existing United States Food and Drug Administration (FDA) licensed vaccines (ACAM2000, JYNNEOS), novel therapeutics, and other public health measures, is of utmost importance.

Materials and methods

Ethics statement

All animal studies were conducted in compliance with the Animal Welfare Act and other federal statutes and regulations relating to animals and experiments involving animals and adheres to principles stated in the Guide for the Care and Use of Laboratory Animals, National Research Council. All animal experimental protocols were approved by a preexisting internal institutional animal care and use committee (IACUC). The facilities where this research was conducted are fully accredited by the Association for Assessment and Accreditation of Laboratory Animal Care International. Animals meeting criteria were humanely euthanized.

Animals, exposure, and treatment

Details about the design, conduct and outcomes of the animals utilized within this manuscript have been previously published by Mucker et al.(23). Briefly, a double-blinded study using 18 male poxvirus naïve Mauritius cynomolgus macaques having age and weight ranging from 2.9 to 6.5 years and 4.8–7.1 kg, respectively, were randomized into two tecovirimat treatment groups and a single placebo control group. All animals were dosed with either placebo or a combination of placebo and tecovirimat for 16 consecutive days (days 2–17). More specifically, animals in the tecovirimat treated groups either received drug on days 2–15 or days 4–17, with placebo administration on days 16 and 17, or 2 and 3, respectively. Animals in the control group received placebo from days 2–17. On Study Day 0, NHPs were intravenously infected with 1 x 10^8 pfu of Harper strain of VARV in one mL volume.

The overall goal of these studies was to evaluate the post-exposure prophylactic (pre-lesion/ rash onset) and therapeutic (post-lesion/rash onset) efficacy of tecovirimat against VARV using the intravenous lesional model of smallpox in nonhuman primates (NHPs). In this model, skin and/or mucosal lesions/rash are typically observed three to five days after exposure. Treatments were delayed by either two (pre-lesion onset) or four (post-lesion onset for most animals) days post-exposure to assess tecovirimat for post exposure prophylaxis and therapeutic indications. Animals were orally administered placebo or tecovirimat and/or placebo daily from Days 2–17 to maintain blinding. More specifically, NHPs received 10 mg/kg/ day of tecovirimat for a total of fourteen days, with placebo given two days prior to initiation (Day 2 treated group), or two days after completion of the tecovirimat regimen (23). The primary endpoint was total lesion count due to smallpox disease. Other experimental endpoints included viral loads (based on qPCR [39]), lesion onset and progression, and survival.

Volume dosing for both test and control articles (e.g., tecovirimat or placebo) were calculated on individual weights of animals at the initiation of the study. All experiments and procedures involving VARV were performed at the Center for Disease Control and Prevention (CDC). Formalin-fixed, paraffin-embedded blocks from this study are stored at USAMRIID in a temperature and humidity-controlled environment.

Gross necropsy

A full necropsy was performed on all animals and recorded by a board-certified veterinary pathologist. A summary of animal numbers, pathology accession numbers, challenge dates, and necropsy dates is presented in Table 1. The following tissues were collected at necropsy and immediately immersed in 10% neutral buffered formalin (NBF): skin (with, if present, and without pox lesions), tongue; tonsil, lip, nares, mandibular, tracheobronchial, axillary, inguinal, and mesenteric lymph nodes, submandibular salivary gland, larynx, esophagus, trachea, thyroid gland, parathyroid gland, testis, liver, spleen, adrenal glands, pituitary gland, kidneys, urinary bladder, prostate gland, heart, lung, skeletal muscle sciatic nerve, brachial plexus, stomach, pylorus, duodenum, jejunum, ileum, cecum, proximal and distal colon, brain, eyes, and femoral bone marrow. Tissues were fixed for a minimum of 21 days in 10% neutral buffered formalin. The in-life portion of this study was completed at the CDC and, after viral inactivation via formalin fixation, FFPE tissues were analyzed by the pathology team at USAMRIID, Fort Detrick, Maryland.

Histopathology

The tissues were trimmed and processed according to standard protocol. Sections were trimmed to $5-6 \mu m$ and routinely stained with hematoxylin and eosin. Replicate sections were processed for immunohistochemistry (IHC), *in situ* hybridization (ISH) and

immunofluorescence (IFA). A board-certified veterinary pathologist evaluated the H&E, IHC, and ISH by light microscopy. IFA was evaluated by a molecular biologist utilizing confocal microscopy. Results for each animal are summarized in Tables 5–9.

Immunohistochemistry

Immunohistochemistry (IHC) was performed on replicate tissue sections using an Envision kit (Dako Agilent Pathology Solutions). An in-house polyclonal antibody against vaccinia virus (#1293, USAMRIID) was used at a dilution of 1:3500. After deparaffinization, rehydration, methanol-hydrogen peroxide blocking, and decrosslinking using higher-pH antigen retrieval buffer, sections were covered with primary antibody and incubated at room temperature for sixty minutes. They were rinsed, and then the peroxidase-labeled polymer (secondary antibody) was applied for thirty minutes. Slides were rinsed and a substrate-chromogen solution was applied for 5 minutes. All slides were exposed to brown chromogenic substrate, 3,3'-diaminobenzidine (DAB; Dako Agilent Pathology Solutions), counterstained with hematoxy-lin, dehydrated, and coverslipped.

In Situ hybridization

In situ hybridization was performed on haired skin and mucosal epithelium. To detect smallpox viral messenger RNA (mRNA) in FFPE tissues, ISH was performed using the RNAscope 2.5 HD Detection Kit (RED) for FFPE Tissues (Advanced Cell Diagnostics, Newark, CA, USA) according to the manufacturer's instructions. Briefly, an ISH probe targeting fragment 2–1311 of smallpox virus-specific gene B19R [13], which aligns to 165795–167421 of LT706528.1., was designed and synthesized by Advanced Cell Diagnostics (Cat# 534681) and an ISH probe targeting MPXV-specific gene D1L was reported previously (Cat#534671, Advanced Cell Diagnostics). Tissue sections were deparaffinized with Xyless II (Val Tech Diagnostics, Brackenridge, PA, USA), underwent a series of ethanol washes and were peroxidase blocking, heated in kit-provided antigen retrieval buffer, and then digested by kit-provided proteinase. Sections were exposed to ISH target probe pairs and incubated at 40°C in a hybridization oven for 2 h. After rinsing, ISH signal was amplified using kit-provided pre-amplifier and amplifier conjugated to alkaline phosphatase and incubated with a Fast Red substrate solution for 10 min at room temperature. Sections were then stained with hematoxylin, air-dried, and coverslipped.

Immunofluorescence

Immunofluorescence was performed on haired skin, and mucosal epithelium. Formalin-fixed paraffin embedded tissue sections were deparaffinized using xylene and a series of ethanol washes. The sections were heated in Tris-EDTA buffer (10mM Tris Base, 1mM EDTA Solution, 0.05% Tween 20, pH 9.0) for 20 minutes to reverse formaldehyde crosslinks. After rinses with phosphate-buffered saline (PBS), pH 7.4 (Thermo Fisher Scientific, Walthan, MA, USA), sections were blocked overnight with PBT (PBS + 0.1% Tween-20) containing 5% normal goat serum (Millipore Sigma, Burlington, MA, USA) at 4°C. Sections were then incubated with the following primary antibodies for 2 h at room temperature as follows: rabbit polyclonal antibody (USAMRIID, Frederick, MD USA) against vaccinia virus, which cross binds to smallpox virus, at a dilution of 1:500 and mouse monoclonal anti-human CD68 antibody (Clone KP1; Dako Agilent Pathology Solutions) at a dilution of 1:200 or mouse monoclonal anti-Cytokeratin 10 antibody (ab20208; Abcam, Cambridge, MA, USA) at a dilution of 1:100. After rinsing in PBT, sections were incubated with secondary goat Alexa Fluor 488-conjugated anti-rabbit and with goat Alexa Fluor 561-conjugated anti-mouse antibody for 1 h at room temperature.

Sections were coverslipped using VECTASHIELD antifade mounting medium with DAPI (Vector Laboratories, Burlingame, CA, USA). Images were captured on a LSM 880 Confocal Microscope (Zeiss, Oberkochen, Germany) and processed using open-source ImageJ software (National Institutes of Health, Bethesda, MD, USA).

Electron microscopy

Three 1 mm tissue punches were made from the six tissue blocks provided. Samples were heated for at 60°C for 45 minutes and deparaffinized in Xyless II for 30 minutes. The 18 tissue plugs were rehydrated in 100% ethanol for 15 minutes and then 95% and 70% ethanol before being fixed in 4% paraformaldehyde and 1% glutaraldehyde and 0.1M sodium cacodylate buffer for 1hour. Samples were subsequently rinsed with 0.2M sucrose in buffer for 30 minutes. Samples were dehydrated with increasing concentration of ethanol (50%, 75%, 95%, and 100%) for 10 minutes each. The samples were further dehydrated in 100% propylene oxide (PO) and 100% ethanol [1:1], followed by two changes of 100% PO for 10 minutes. Samples were incubated in 2 exchanges of pure epoxy resin before being embedded, oriented and polymerized at 60°C for 48 hours.

Five hundred-nanometer sections were cut and counterstained with toluidine blue from all 18 samples. Toluidine blue stained images verified tissue integrity and served as a guide for TEM imaging. 75–85 nm thin sections were cut of these regions, collected on 200 mesh copper grids and post-section stained with uranyl acetate and Reynold's lead citrate. Grids were imaged on the JEOL TEM1400 at various magnifications using the Gatan Rio9 camera.

Acknowledgments

We would like to thank the USAMRIID pathology department for their help in this endeavor. We would like to also thank the Poxvirus and Rabies Branch at the CDC for supporting the nonhuman primate study. Specifically, we would like to thank Victoria Olson, Cody Clemons, Nadia Gallardo, Jason Mehal, Robert Holman, Darin Carroll and Paul Hudson, who contributed to the original nonhuman primate studies that lead to the current research.

Declarations

The views, opinions, and/or findings contained herein are those of the authors and should not be construed as an official Department of Army position, policy, or decision unless so designated by other documentation.

Author Contributions

- **Conceptualization:** Todd M. Bell, Joshua D. Shamblin, Douglas W. Grosenbach, Dennis E. Hruby, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.
- **Data curation:** Todd M. Bell, Paul Facemire, Jeremy J. Bearss, Jo Lynne Raymond, Jennifer Chapman, Xiankun Zeng, Janice A. Williams, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.
- Formal analysis: Todd M. Bell, Paul Facemire, Jeremy J. Bearss, Jo Lynne Raymond, Jennifer Chapman, Xiankun Zeng, Janice A. Williams, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.

Funding acquisition: Douglas W. Grosenbach, Dennis E. Hruby.

- **Investigation:** Todd M. Bell, Paul Facemire, Xiankun Zeng, Joshua D. Shamblin, Douglas W. Grosenbach, Dennis E. Hruby, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.
- **Methodology:** Todd M. Bell, Paul Facemire, Xiankun Zeng, Joshua D. Shamblin, Douglas W. Grosenbach, Dennis E. Hruby, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.
- **Project administration:** Todd M. Bell, Jennifer Chapman, Douglas W. Grosenbach, Dennis E. Hruby, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.
- Resources: Douglas W. Grosenbach, Dennis E. Hruby, Arthur J. Goff.
- Supervision: Paul Facemire, Jennifer Chapman, Joshua D. Shamblin, Douglas W. Grosenbach, Inger K. Damon, Eric M. Mucker.
- Validation: Joshua D. Shamblin.

Visualization: Todd M. Bell, Joshua D. Shamblin, Janice A. Williams, Eric M. Mucker.

- Writing original draft: Todd M. Bell, Paul Facemire, Jennifer Chapman, Xiankun Zeng, Joshua D. Shamblin, Janice A. Williams, Eric M. Mucker.
- Writing review & editing: Todd M. Bell, Paul Facemire, Jeremy J. Bearss, Jo Lynne Raymond, Jennifer Chapman, Xiankun Zeng, Joshua D. Shamblin, Janice A. Williams, Douglas W. Grosenbach, Dennis E. Hruby, Inger K. Damon, Arthur J. Goff, Eric M. Mucker.

References

- Fenner F. Smallpox: emergence, global spread, and eradication. Hist Philos Life Sci. 1993; 15(3):397– 420. PMID: 7529932
- 2. Henderson DA. Smallpox eradication. Public Health Rep. 1980; 95(5):422–6. PMID: 7422808
- Henderson DA. Principles and lessons from the smallpox eradication programme. Bull World Health Organ. 1987; 65(4):535–46. PMID: <u>3319270</u>
- Nature Fenner F., nurture and my experience with smallpox eradication. Med J Aust. 1999; 171(11– 12):638–41.
- 5. Long CM, Marzi A. Biodefence research two decades on: worth the investment? Lancet Infect Dis. 2021; 21(8):e222–e33. https://doi.org/10.1016/S1473-3099(21)00382-0 PMID: 34331891
- Sell TK, Watson M. Federal agency biodefense funding, FY2013-FY2014. Biosecur Bioterror. 2013; 11 (3):196–216. https://doi.org/10.1089/bsp.2013.0047 PMID: 23906009
- Gursky E, Inglesby TV, O'Toole T. Anthrax 2001: observations on the medical and public health response. Biosecur Bioterror. 2003; 1(2):97–110. https://doi.org/10.1089/153871303766275763 PMID: 15040187
- Watson CR, Watson M, Sell TK. Public Health Preparedness Funding: Key Programs and Trends From 2001 to 2017. Am J Public Health. 2017; 107(S2):S165–S7. <u>https://doi.org/10.2105/AJPH.2017</u>. 303963 PMID: 28892451
- Breman JG, Henderson DA. Poxvirus dilemmas—monkeypox, smallpox, and biologic terrorism. N Engl J Med. 1998; 339(8):556–9. https://doi.org/10.1056/NEJM199808203390811 PMID: 9709051
- Simon JD. Biological terrorism. Preparing to meet the threat. JAMA. 1997; 278(5):428–30. https://doi. org/10.1001/jama.278.5.428 PMID: 9244336
- Roberts R, Styrt B, McCune S. FDA perspective on antivirals against biothreats: communicate early and often. Antiviral Res. 2008; 78(1):60–3. https://doi.org/10.1016/j.antiviral.2007.10.001 PMID: 18037169
- Rotz LD, Khan AS, Lillibridge SR, Ostroff SM, Hughes JM. Public health assessment of potential biological terrorism agents. Emerg Infect Dis. 2002; 8(2):225–30. https://doi.org/10.3201/eid0802.010164 PMID: 11897082
- Merchlinsky M, Albright A, Olson V, Schiltz H, Merkeley T, Hughes C, et al. The development and approval of tecoviromat (TPOXX((R))), the first antiviral against smallpox. Antiviral Res. 2019; 168:168–74.

- Bolken TC, Hruby DE. Discovery and development of antiviral drugs for biodefense: experience of a small biotechnology company. Antiviral Res. 2008; 77(1):1–5. https://doi.org/10.1016/j.antiviral.2007. 07.003 PMID: 17765333
- Snoy PJ. Establishing efficacy of human products using animals: the US food and drug administration's "animal rule". Vet Pathol. 2010; 47(5):774–8. <u>https://doi.org/10.1177/0300985810372506</u> PMID: 20551476
- Food, Drug Administration HHS. New drug and biological drug products; evidence needed to demonstrate effectiveness of new drugs when human efficacy studies are not ethical or feasible. Final rule. Fed Regist. 2002; 67(105):37988–98.
- Earl PL, Americo JL, Wyatt LS, Eller LA, Whitbeck JC, Cohen GH, et al. Immunogenicity of a highly attenuated MVA smallpox vaccine and protection against monkeypox. Nature. 2004; 428(6979):182–5. https://doi.org/10.1038/nature02331 PMID: 15014500
- DeLaurentis CE, Kiser J, Zucker J. New Perspectives on Antimicrobial Agents: Tecovirimat for Treatment of Human Monkeypox Virus. Antimicrob Agents Chemother. 2022; 66(12):e0122622. https://doi. org/10.1128/aac.01226-22 PMID: 36374026
- Warner BM, Klassen L, Sloan A, Deschambault Y, Soule G, Banadyga L, et al. In vitro and in vivo efficacy of tecovirimat against a recently emerged 2022 monkeypox virus isolate. Sci Transl Med. 2022; 14 (673):eade7646. https://doi.org/10.1126/scitranslmed.ade7646 PMID: 36318038
- Amantana A, Chen Y, Tyavanagimatt SR, Jones KF, Jordan R, Chinsangaram J, et al. Pharmacokinetics and interspecies allometric scaling of ST-246, an oral antiviral therapeutic for treatment of orthopoxvirus infection. PLoS One. 2013; 8(4):e61514. <u>https://doi.org/10.1371/journal.pone.0061514</u> PMID: 23637845
- Berhanu A, Prigge JT, Silvera PM, Honeychurch KM, Hruby DE, Grosenbach DW. Treatment with the smallpox antiviral tecovirimat (ST-246) alone or in combination with ACAM2000 vaccination is effective as a postsymptomatic therapy for monkeypox virus infection. Antimicrob Agents Chemother. 2015; 59 (7):4296–300. https://doi.org/10.1128/AAC.00208-15 PMID: 25896687
- Duraffour S, Lorenzo MM, Zoller G, Topalis D, Grosenbach D, Hruby DE, et al. ST-246 is a key antiviral to inhibit the viral F13L phospholipase, one of the essential proteins for orthopoxvirus wrapping. J Antimicrob Chemother. 2015; 70(5):1367–80. https://doi.org/10.1093/jac/dku545 PMID: 25630650
- Mucker EM, Goff AJ, Shamblin JD, Grosenbach DW, Damon IK, Mehal JM, et al. Efficacy of tecovirimat (ST-246) in nonhuman primates infected with variola virus (Smallpox). Antimicrob Agents Chemother. 2013; 57(12):6246–53. https://doi.org/10.1128/AAC.00977-13 PMID: 24100494
- Zaitseva M, Shotwell E, Scott J, Cruz S, King LR, Manischewitz J, et al. Effects of postchallenge administration of ST-246 on dissemination of IHD-J-Luc vaccinia virus in normal mice and in immune-deficient mice reconstituted with T cells. J Virol. 2013; 87(10):5564–76. <u>https://doi.org/10.1128/JVI.03426-12</u> PMID: 23468500
- Leeds JM, Fenneteau F, Gosselin NH, Mouksassi MS, Kassir N, Marier JF, et al. Pharmacokinetic and pharmacodynamic modeling to determine the dose of ST-246 to protect against smallpox in humans. Antimicrob Agents Chemother. 2013; 57(3):1136–43. https://doi.org/10.1128/AAC.00959-12 PMID: 23254433
- Grosenbach DW, Jordan R, Hruby DE. Development of the small-molecule antiviral ST-246 as a smallpox therapeutic. Future Virol. 2011; 6(5):653–71. https://doi.org/10.2217/fvl.11.27 PMID: 21837250
- Cashman KA, Wilkinson ER, Zeng X, Cardile AP, Facemire PR, Bell TM, et al. Immune-Mediated Systemic Vasculitis as the Proposed Cause of Sudden-Onset Sensorineural Hearing Loss following Lassa Virus Exposure in Cynomolgus Macaques. mBio. 2018; 9(5). <u>https://doi.org/10.1128/mBio.01896-18</u> PMID: 30377282
- Liu J, Coffin KM, Johnston SC, Babka AM, Bell TM, Long SY, et al. Nipah virus persists in the brains of nonhuman primate survivors. JCI Insight. 2019; 4(14). https://doi.org/10.1172/jci.insight.129629 PMID: 31341108
- Liu J, Mucker EM, Chapman JL, Babka AM, Gordon JM, Bryan AV, et al. Retrospective detection of monkeypox virus in the testes of nonhuman primate survivors. Nat Microbiol. 2022; 7(12):1980–6. https://doi.org/10.1038/s41564-022-01259-w PMID: 36253513
- Liu J, Trefry JC, Babka AM, Schellhase CW, Coffin KM, Williams JA, et al. Ebola virus persistence and disease recrudescence in the brains of antibody-treated nonhuman primate survivors. Sci Transl Med. 2022; 14(631):eabi5229. https://doi.org/10.1126/scitranslmed.abi5229 PMID: 35138912
- Zeng X, Blancett CD, Koistinen KA, Schellhase CW, Bearss JJ, Radoshitzky SR, et al. Identification and pathological characterization of persistent asymptomatic Ebola virus infection in rhesus monkeys. Nat Microbiol. 2017; 2:17113. https://doi.org/10.1038/nmicrobiol.2017.113 PMID: 28715405
- 32. Smith DR, Shoemaker CJ, Zeng X, Garrison AR, Golden JW, Schellhase CW, et al. Persistent Crimean-Congo hemorrhagic fever virus infection in the testes and within granulomas of non-human

primates with latent tuberculosis. PLoS Pathog. 2019; 15(9):e1008050. https://doi.org/10.1371/journal. ppat.1008050 PMID: 31557262

- 33. Wahl-Jensen V, Cann JA, Rubins KH, Huggins JW, Fisher RW, Johnson AJ, et al. Progression of pathogenic events in cynomolgus macaques infected with variola virus. PLoS One. 2011; 6(10):e24832. https://doi.org/10.1371/journal.pone.0024832 PMID: 21998632
- Begum JPS, Ngangom L, Semwal P, Painuli S, Sharma R, Gupta A. Emergence of monkeypox: a worldwide public health crisis. Hum Cell. 2023:1–17. https://doi.org/10.1007/s13577-023-00870-1 PMID: 36749539
- 35. Bragazzi NL, Kong JD, Mahroum N, Tsigalou C, Khamisy-Farah R, Converti M, et al. Epidemiological trends and clinical features of the ongoing monkeypox epidemic: A preliminary pooled data analysis and literature review. J Med Virol. 2023; 95(1):e27931. https://doi.org/10.1002/jmv.27931 PMID: 35692117
- Meo SA, Klonoff DC. Human monkeypox outbreak: global prevalence and biological, epidemiological and clinical characteristics—observational analysis between 1970–2022. Eur Rev Med Pharmacol Sci. 2022; 26(15):5624–32. https://doi.org/10.26355/eurrev_202208_29436 PMID: 35993662
- Grosenbach DW, Honeychurch K, Rose EA, Chinsangaram J, Frimm A, Maiti B, et al. Oral Tecovirimat for the Treatment of Smallpox. N Engl J Med. 2018; 379(1):44–53. https://doi.org/10.1056/ NEJMoa1705688 PMID: 29972742
- Jordan R, Goff A, Frimm A, Corrado ML, Hensley LE, Byrd CM, et al. ST-246 antiviral efficacy in a nonhuman primate monkeypox model: determination of the minimal effective dose and human dose justification. Antimicrob Agents Chemother. 2009; 53(5):1817–22. <u>https://doi.org/10.1128/AAC.01596-08</u> PMID: 19223621
- Mucker EM, Hartmann C, Hering D, Giles W, Miller D, Fisher R, et al. Validation of a pan-orthopox realtime PCR assay for the detection and quantification of viral genomes from nonhuman primate blood. Virol J. 2017; 14(1):210. https://doi.org/10.1186/s12985-017-0880-8 PMID: 29100534